Author's response to reviews

Title: Idiopathic Pleuroparenchymal Fibroelastosis: Consideration of a Clinicopathological Entity in a Series of Japanese Patients

Authors:

Hideki Kusagaya (kusagaya0726@gmail.com)
Yutaro Nakamura (nakayuta@hama-med.ac.jp)
Masato Kono (konomasato28@yahoo.co.jp)
Yusuke Kaida (ykaida@gmail.com)
Shigeki Kuroishi (kuroishi@vmail.plala.or.jp)
Noriyuki Enomoto (norieno@hama-med.ac.jp)
Tomoyuki Fujisawa (fujisawa@hama-med.ac.jp)
Koshi Yokomura (yo.koshi@sis.seirei.or.jp)
Naoki Inui (inui@hama-med.ac.jp)
Takafumi Suda (suda@hama-med.ac.jp)
Thomas V Colby (colby.thomas@mayo.edu)
Kingo Chida (chidak11@hama-med.ac.jp)
Naoki Koshimizu (pineapplefish1219@yahoo.co.jp)

Version: 5 Date: 28 November 2012

Author's response to reviews: see over
Dear Dr. Timothy Shipley
Executive Editor
BMC pulmonary medicine

Thank you very much for your letter of November 22, 2012, with regard to our manuscript (MS: 1207584320740548) together with the comments from the reviewer. The comments of the reviewer have been helpful in allowing us to improve our manuscript. We hope that the manuscripts now acceptable for publication in BMC Pulmonary Medicine.

Sincerely,
Yutaro Nakamura

RESPONSE TO REVIEWER #1 (Dr. Matthew Jankowich)

Thank you for pointing out our wording issue. We sincerely corrected, according to reviewer’s suggestion. And there was other misspelling. We corrected to “Written informed consent” from “Witten informed consent” in the Consent. Thank you again.